Author's response to reviews

Title: The GFR estimated by different new and old equations as a predictor of important outcomes in elderly patients

Authors:

Gijs Van Pottelbergh (gijs.vanpottelbergh@med.kuleuven.be)
Bert Vaes (bert.vaes@med.kuleuven.be)
Cathy Matheï (cathy.mathei@med.kuleuven.be)
Delphine Legrand (delphine.legrand@uclouvain.be)
Wim Adriaensen (wim.adriaensen@med.kuleuven.be)
Pierre Wallemacq (pierre.wallemacq@med.kuleuven.be)
Jean-Marie Degryse (jan.degryme@med.kuleuven.be)

Version: 3 Date: 10 January 2014

Author's response to reviews:

dear editor,

we have made the requested changes:

1. we changed the sentence towards "The BIS equation predicts less accurately renal deaths."

2. we corrected the incorrect notifications of cystatin C

kind regards